Author's response to reviews

Title: Vitamin A status between 6 months and 2 years of age in Guinea-Bissau

Authors:

Niels Danneskiold-Samsøe (nds@bio.ku.dk)
Ane B Fisker (a.fisker@bandim.org)
Mathias J Jørgensen (mathiasjul@gmail.com)
Henrik Ravn (HJN@ssi.dk)
Andreas Andersen (a.andersen@bandim.org)
Ibraima D Balde (i.balde@bandim.org)
Christian Leo-Hansen (chr_leohansen@hotmail.com)
Amabelia Rodrigues (a.rodrigues@bandim.org)
Peter Aaby (p.aaby@bandim.org)
Christine S Benn (cb@ssi.dk)

Version: 2 Date: 4 June 2012

Author's response to reviews:

- statement about informed consent for participation in the study added to methods section
- specific names of the organizations that granted ethical approval to the study going ahead added
- Statement about Helsinki Declaration added to methods section
- cover letter removed
- trial registration number added to end of abstract